**Author’s response to reviews**

**Title:** How to squat? Effect of various stance widths, foot placement angles and level of experience on knee, hip and trunk motion and loading

**Authors:**

Silvio Lorenzetti (slorenzetti@ethz.ch)
Mira Ostermann (mira.ostermann@gmail.com)
Fabian Zeidler (fabian.zeidler@hest.ethz.ch)
Pia Zimmer (pia.zimmer@hest.ethz.ch)
Lina Jentsch (jentschl@student.ethz.ch)
Renate List (rl@ethz.ch)
William R. Taylor (taylorb@ethz.ch)
Florian Schellenberg (fs@ethz.ch)

**Version:** 4  **Date:** 11 Jun 2018

**Author’s response to reviews:**

1) Highlights: please remove this text as this is not part of the journal format

This text has been removed.

2) Line 450: in the sentence the 'The participant...', please change on to in. i.e. The participant in ....

"in" has been removed, "on" has been included.

3) Authors contributions: did WRT play any further role in the study beyond supporting the text and writing part of the manuscript as this contribution is not sufficient to warrant authorship. Please see our Editorial Policies at [https://www.biomedcentral.com/getpublished/editorial-policies#authorship](https://www.biomedcentral.com/getpublished/editorial-policies#authorship) for further information.

The contribution of WRT has been changed to:"WRT: helped with analysis, supported the texting and wrote part of the manuscript. "